Author’s response to reviews

Title: The impact of music therapy on managing neuropsychiatric symptoms for people with dementia and their carers: a randomised controlled feasibility study

Authors:

Ming Hung Hsu (ming.hunghsu@mha.org.uk)
Rosamund Flowerdew (ros.flowerfew@mha.org.uk)
Michael Parker (michael.parker@anglia.ac.uk)
Jörg Fachner (jorg.fachner@anglia.ac.uk)
Helen Odell-Miller (helen.odell-miller@anglia.ac.uk)

Version: 3
Date: 16 January 2015

Author’s response to reviews: see over
Changes to the manuscript have been made as per CONSORT guidelines and are outlined as follows:

- Method, p7, lines 168-170. Text added to ‘Setting’.
- Method, p13, lines 290, 293-294. Text added to ‘Sample size’.
- Method, pp13-14, lines 298-304. Text added to ‘Randomisation’.
- Results, p15, lines 335, 340, 343-344. Text added to ‘Recruitment’.
- Results, p17, lines 392-395. Text added to ‘Numbers analysed’.
- Results, p22, lines 506-507. Text added to ‘Harms’.
- Consort check list included.
- Flowchart included as Figure 1.